Author's response to reviews

Title: One-year follow-up of patients with long-lasting post-herpetic neuralgia.

Authors:

Francesca Pica (pica@uniroma2.it)
Antonio Gatti (gatti@med.uniroma2.it)
Marco Divizia (ospedalesenzadolore@ptvonline.it)
Marzia Lazzari (marzialaz@hotmail.it)
Marco Ciotti (marco.ciotti@ptvonline.it)
Alessandro Fabrizio Sabato (sabato@med.uniroma2.it)
Antonio Volpi (volpi@med.uniroma2.it)

Version: 2
Date: 8 November 2013

Author's response to reviews:

Dear Prof Volpi

Thank you for your recent submission to BMC Medicine (referenced above).

My colleagues and I have carefully considered your manuscript and I'm very sorry to say it seems to us that it would not be appropriate for BMC Medicine. It is our feeling that the manuscript does not represent a sufficiently profound step forward in understanding to be of broad interest to researchers outside the field.

However, we feel that your manuscript may be suited to the scope and readership of another of our journals, BMC Infectious Diseases (http://www.biomedcentral.com/bmcinfectdis)

We have copied over the details of your manuscript (your submission record and files) to BMC Infectious Diseases and if you wish to pursue submission in this journal please do so by logging on to 'My Manuscripts' at http://www.biomedcentral.com/bmcmed/my/manuscripts

Benefits of transferring your submission of this manuscript to BMC Infectious Diseases may include:
* easier publication of your work, saving time finding and submitting to an alternative publisher;
* faster publication, we will transfer your manuscript record and reviewer comments to the suggested journal for you;
* reaching the right audience for your work, we believe the suggested journal may be a better fit.

Please note that BMC Infectious Diseases is editorially independent from BMC Medicine and therefore the manuscript will be assessed by the BMC Infectious Diseases editorial team as a new submission.
Prior to submitting to BMC Infectious Diseases, we would draw your attention to the article processing charge for manuscripts published in BMC Infectious Diseases, which is £1230/US$1885/€1475.

Thank you for your interest in BMC Medicine, and we wish you success with your submission to BMC Infectious Diseases.

With best wishes,

Joanna Denyer, PhD
Assistant Editor
BMC Medicine